Author's response to reviews

Title: PARP inhibition and the radiosensitizing effects of the PARP inhibitor ABT-888 in in vitro hepatocellular carcinoma models

Authors:

Clément Guillot (clementguillot84@gmail.com)
Vincent Favaudon (vincent.favaudon@curie.fr)
Zdenko Herceg (herceg@iarc.fr)
Charlotte Sagne (charlotte.sagne@gmail.com)
Sylvie Sauvaigo (sylvie.sauvaigo@lxrepair.com)
Philippe Merle (philippe.merle@inserm.fr)
Janet Hall (janet.hall@curie.fr)
Isabelle Chemin (isabelle.chemin@inserm.fr)

Version: 4 Date: 1 August 2014

Author's response to reviews:

Dear Madam, Sir

Please find the response to the editors request for our paper entitled “PARP inhibition and the radiosensitizing effects of the PARP inhibitor ABT-888 in in vitro hepatocellular carcinoma models” by Guillot et al.

EDITORIAL REQUEST:

Please could you clarify the full name of the local ethics committee which approved the use of the human samples.

The full name and contact details of the local ethics committee which approved the use of the human samples are:

CPP Sud-Est IV
Centre Léon Bérard - 28, rue Laënnec - 69373 LYON cedex 8
FRANCE
Phone number: 00 33 4 78 78 27 61
e-mail contact: odette.maritaz@lyon.unicancer.fr

We included the name of the local ethics committee in the methods section of the manuscript.

Best regards,

Lyon 08.01.14

Dr Isabelle Chemin
Cancer Research Center of Lyon
Leader Team 16, Hepatocarcinogenesis and viral infection